Author's response to reviews

Title: Impact of ambient air pollution on gestational age is modified by season in Sydney, Australia

Authors:

Bin Jalaludin (b.jalaludin@unsw.edu.au)
Trish Mannes (trish.mannes@email.cs.nsw.gov.au)
Geoffrey Morgan (gmorg@nrahs.nsw.gov.au)
Doug Lincoln (dougl@nrahs.nsw.gov.au)
Vicky Sheppeard (yshep@doh.health.nsw.gov.au)
Stephen Corbett (stephen_corbett@wsahs.nsw.gov.au)

Version: 3 Date: 21 May 2007

Author's response to reviews: see over
18 May 2007

Dr D Ozonoff
The Editor
Environmental Health

Dear Dr Ozonoff

Re: Further revisions to manuscript titled “Impact of ambient air pollution on gestational age is modified by season in Sydney, Australia”

Thank you for accepting our manuscript for publication following our responses to the Reviewers’ comments.

We have now also responded to the minor corrections to the references (bolding the volumes) and tables (shortening the titles of tables and presenting the tables in portrait format) that had been requested. We have also added the internet address of the report cited in reference #23 although we did not access the internet for the report as we already had a hard copy of the report. However, providing the internet address will enable the Journal readers to have easy access to the report. In our copy of the submitted manuscript, we did not find that the first sentence of the Discussion section was missing, and therefore we have not made any changes there.

In our revised manuscript we had highlighted our responses to the major and minor essential revisions as suggested by the reviewers. As the revisions have been accepted, we have removed the highlights from this version of the manuscript.

We will be happy to respond to any further comments from the Editors or the Reviewers.

Yours sincerely

A/Prof Bin Jalaludin
Centre for Research, Evidence Management and Surveillance